Author's response to reviews

Title: Design and baseline characteristics of a prospective cohort study for determinants of osteoporotic fracture in community-dwelling elderly Japanese men: the Fujiwara-kyo Osteoporosis Risk in Men (FORMEN) Study

Authors:

Masayuki Iki (masa@med.kindai.ac.jp)
Yuki Fujita (yfujita@med.kindai.ac.jp)
Junko Tamaki (jtamaki@med.kindai.ac.jp)
Katsuyasu Kouda (kouda@med.kindai.ac.jp)
Akiko Yura (ayuko@med.kindai.ac.jp)
Eiko Kadowaki (eiko@med.kindai.ac.jp)
Yuho Sato (y-sato@jindai.ac.jp)
Jong-Seong Moon (mjs@tgu.ac.jp)
Nozomi Okamoto (onozomi@nmu-gw.naramed-u.ac.jp)
Norio Kurumatani (knorio@nmu-gw.cc.naramed-u.ac.jp)
Study Group for Functioning Capacity and Quality of Life in Elderly Japanese (Fujiwara-kyo Study Group) (knorio@nmu-gw.cc.naramed-u.ac.jp)

Version: 10 Date: 16 December 2009

Author's response to reviews: see over
Dr. Melissa Norton
Editor-in-chief
BMC Musculoskeletal Disorders

Dec 16, 2009

Dear Dr. Norton,

We have revised again the manuscript below according to formatting requirements from the BioMed Central Editorial Production Team.

Authors: Masayuki Iki, Yuki Fujita, Junko Tamaki, Katsuyasu Kouda, Akiko Yura, Eiko Kadowaki, Yuho Sato, Jong-Seong Moon, Nozomi Okamoto, Norio Kurumatani, and Fujiwara-kyo Study Group
Title: Design and baseline characteristics of a prospective cohort study for determinants of osteoporotic fracture in community-dwelling elderly Japanese men: the Fujiwara-kyo Osteoporosis Risk in Men (FORMEN) Study
Journal: BMC Musculoskeletal Disorders
MS: 4770355230283952

The following minor revisions have been made.
1. We found several typographical errors in the previous version of manuscript and corrected them. We believe no error in this final version.
2. We added some lines in Table 2.

I would be pleased if you find this version of manuscript satisfactory for publication in BMC Musculoskeletal Disorders. I look forward to hearing from you, soon.

Yours sincerely,

Masayuki Iki, MD, PhD, Professor
Department of Public Health
Kinki University School of Medicine
377-2 Oono-higashi
Osaka-Sayama, Osaka 589-8511, Japan
Tel  +81-72-366-0221 Ext. 3270
Fax  +81-72-367-8262
e-mail  masa@med.kindai.ac.jp